Author's response to reviews

Title: The problem of obesity among adolescents in Hong Kong: a comparison using various diagnostic criteria

Authors:

Gary T Ko (gtc_ko@yahoo.com.hk)
Risa Ozaki (risaozaki@yahoo.com)
Gary W Wong (gwong@cuhk.edu.hk)
Alice P Kong (alicekong@cuhk.edu.hk)
Wing-Yee So (wingyeeso@cuhk.edu.hk)
Peter C Tong (ptong@cuhk.edu.hk)
Michael H Chan (chanhm@ha.org.hk)
Chung-Shun Ho (hocs@ha.org.hk)
Christopher W Lam (cwklam@cuhk.edu.hk)
Juliana C Chan (jchan@cuhk.edu.hk)

Version: 4 Date: 19 February 2008

Author's response to reviews:

19 Feb 2008
Editor-in-Chief
BMC Pediatrics

Dear Sir

Re: MS: 4147174341530927: The Problem of Obesity Among Adolescents in Hong Kong: A Comparison Using Various Diagnostic Criteria (previously: The Problem of Childhood Obesity in Hong Kong - A Cross-Sectional Observation Study)

Thank you very much for your email on 13 Feb 2008. We are grateful that you and your Editorial Board have accepted our paper. We have formatted the manuscript according to your instruction. We hope the letter is now up to your expectation.

Thank you for your attention.

Yours faithfully,

(signed)____________________
Gary TC Ko, for all authors